Introduction

About CPSC 097: Senior Conference

This course provides honors and course majors an opportunity to delve more deeply into a particular topic in computer science, synthesizing material from previous courses. Topics have included advanced algorithms, networking, evolutionary computation, complexity, encryption and compression, and parallel processing. CPSC 097 is the usual method used to satisfy the comprehensive requirement for a computer science major.

During the 2002-2003 academic year, the Senior Conference was led by Richard Wicentowski in the area of Natural Language Processing.

Computer Science Department

Charles Kelemen, Edward Hicks Magill Professor and Chair Lisa Meeden, Associate Professor Tia Newhall, Assistant Professor Richard Wicentowski, Assistant Professor Ali Erkan, Visiting Assistant Professor

Program Committee Members

Mark Angelillo Haw-Bin Chai Julie Corder Hollis Easter Sean Finney Kuzman Ganchev Todd Gillette Feng He Nori Heikkinen Oliver Hsu Yoshi Komori J. McConnell Shannon McGrael Erik Osheim Sven Olsen Jeff Regier Ben Schak Mike Smith Michael Spiegel Daniel Sproul Andrew Stout Yeelin Tan Anteneh Tesfaye Pascal Troemel Jonah Volk Richard Wicentowski Andy Zuppann

Conference Website

http://www.cs.swarthmore.edu/~richardw/cs97-s03/